It is illegal to post this copyrighted PDF on any website. Generalizability of Pharmacologic and Psychotherapy Clinical Trial Results for Posttraumatic Stress Disorder to Community Samples

Silvia Franco, MD^{a,*}; Nicolas Hoertel, MD, MPH^{a,b,c}; Kibby McMahon, BA^{a,d}; Shuai Wang, PhD^a; Jorge Mario Rodríguez-Fernández, MD^a; Hugo Peyre, MD, MPH^e; Frédéric Limosin, MD, PhD^{b,c}; and Carlos Blanco, MD, PhD^a

ABSTRACT

Objective: The present study sought to quantify the generalizability of pharmacologic and psychotherapy clinical trial results in individuals with a *DSM-IV* diagnosis of posttraumatic stress disorder (PTSD) to a large representative community sample.

Methods: Data were derived from the 2004–2005 National Epidemiologic Survey on Alcohol and Related Conditions (NESARC), a large nationally representative sample of the adult US population. We applied a standard set of exclusion criteria representative of pharmacologic and psychotherapy clinical trials to all adults with a *DSM-IV* diagnosis of PTSD in the previous 12 months (n=1,715) and then to a subsample of participants seeking treatment (n=366). Our aim was to assess how many participants with PTSD would fulfill typical eligibility criteria.

Results: We found that more than 6 of 10 respondents from the overall PTSD sample and more than 7 of 10 respondents seeking treatment for PTSD would have been excluded by 1 exclusion criterion or more in a typical pharmacologic trial. In contrast, about 2 of 10 participants in the full sample and about 3 of 10 participants seeking treatment for PTSD would have been excluded in a typical psychotherapy efficacy trial.

Conclusions: We found that psychotherapy trial results may be applied to most patients with PTSD in routine clinical practice. The designers of pharmacologic clinical trials should carefully consider the trade-offs between the application of each exclusion criterion and its impact on representativeness. Specification a priori of the goals of the study, better justification for each exclusion criterion, and estimation of the proportion of individuals ineligible for the trial would assist study design. Developing integrated forms of pharmacotherapy and psychotherapy that simultaneously target commonly overlapping psychiatric disorders may yield more informative results for mental health care providers and research funding agencies.

J Clin Psychiatry 2016;77(8):e975–e981 dx.doi.org/10.4088/JCP.15m10060 © Copyright 2016 Physicians Postgraduate Press, Inc.

^aDepartment of Psychiatry, New York State Psychiatric Institute/Columbia University, New York

^bAssistance Publique-Hôpitaux de Paris (APHP), Corentin Celton Hospital, Department of Psychiatry, Issy-les-Moulineaux; Paris Descartes University, PRES Sorbonne Paris Cité, Paris, France

^cINSERM UMR 894, Psychiatry and Neurosciences Center; Paris Descartes University, PRES Sorbonne Paris Cité, Paris, France

^dDuke University Medical Center, Durham, North Carolina

^eChild and Adolescent Psychiatry Department, Robert Debré Hospital, Paris, France

*Corresponding author: Silvia Franco, MD, Department of Psychiatry, New York State Psychiatric Institute/Columbia University, 1051 Riverside Dr, Unit 69, New York, NY 10032 (corsosi@nyspi.columbia.edu).

andomized controlled clinical trials (RCTs) are often considered the "gold standard" for assessing the efficacy of pharmacologic and psychotherapeutic interventions in clinical research.¹ RCTs apply restrictive eligibility criteria in hope of decreasing heterogeneity in response to treatment (and thereby increasing statistical power), reducing study costs (eg, by excluding participants who are likely to drop out prematurely), ensuring safety of participants (eg, excluding those who could be harmed in the study), and complying with guidelines by regulatory agencies.² However, the high internal validity of clinical trials may be achieved at the cost of diminished external validity (ie, applicability of clinical trial results to routine clinical care) because a substantial proportion of treatment-seeking individuals who are excluded from trials may respond differently to the studied intervention.³ Therefore, there have been calls³⁻⁵ to quantify the generalizability of clinical trials to the broader target population to help frontline practitioners determine the relevance of these studies for their patients and to assist research funding agencies identify gaps in knowledge.

Previous studies have estimated the impact of exclusion criteria on the generalizability of clinical trials for several psychiatric disorders, including bipolar disorder,⁶ panic disorder,⁷ major depressive disorder (MDD),⁸⁻¹² generalized anxiety disorder,¹³ schizophrenia,¹⁴⁻¹⁷ borderline personality disorder,¹ social anxiety disorder,¹⁸ alcohol dependence,^{19,20} nicotine dependence,²¹ and cannabis dependence.²² These studies concluded that the rate of exclusion would range from 50.5% to 87.8%, suggesting limited applicability of clinical trial results to patients seen in routine clinical settings.

Despite the important personal suffering and societal burden associated with posttraumatic stress disorder (PTSD) and the need for more effective treatments,^{23–26} no study to our knowledge has examined the representativeness of pharmacologic clinical trials for PTSD, and only one²⁷ has focused on the generalizability of psychotherapy clinical trials. Given the high prevalence of psychiatric and medical comorbidities associated with PTSD,^{28–34} it is important to examine the generalizability of pharmacologic and psychotherapy trials for PTSD to evaluate the relevance of applying their results to patients in day-to-day clinical practice and guide eligibility criteria selection of future clinical trials.

In the present report, we use the population generalizability estimator method,^{10,20} which applies exclusion criteria commonly used in pharmacologic and psychotherapy clinical

For reprints or permissions, contact permissions@psychiatrist.com. ♦ © 2016 Copyright Physicians Postgraduate Press, Inc. J Clin Psychiatry 77:8, August 2016 PSYCHIATRIST.COM ■ e975 inical Points

It is illegal to post this copyrighted PDF on any website.

- Exclusion criteria applied in pharmacologic clinical trials for PTSD exclude a majority of participants, particularly those seeking treatment, limiting the generalizability of the trial results.
 - Psychotherapy trial results may be applied to most patients with PTSD in routine clinical practice.
- Clinicians and researchers should carefully consider eligibility criteria and their impact on the representativeness of clinical trials for PTSD.

trials for PTSD to a nationally representative sample to assess the generalizability of the criteria to community adults with PTSD.

METHODS

Sample

Data were drawn from the 2004-2005 National Epidemiologic Survey on Alcohol and Related Conditions (NESARC), the second wave that follows upon the Wave 1 NESARC, which was conducted in 2001-2002 and described in detail elsewhere.35,36 The Wave 1 NESARC is a nationally representative face-to-face survey of 43,093 civilian noninstitutionalized US residents aged 18 years and older conducted by the US Census Bureau under the direction of the National Institute on Alcohol Abuse and Alcoholism (NIAAA). The cumulative response rate at Wave 2 was 70.2%, reflecting 34,653 who completed interviews in both Waves. The Wave 2 NESARC data were weighted to reflect design characteristics of the NESARC survey, to adjust for Wave 2 nonresponse, to account for oversampling, and to be representative of the US population.³⁷ The research protocol, including informed consent procedures, received full human subjects review and approval from the US Census Bureau and US Office of Management and Budget.

Assessment of PTSD

Diagnoses were made according to the criteria of the *Diagnostic and Statistical Manual of Mental Disorders*, Fourth Edition (*DSM-IV*) using the NIAAA Alcohol Use Disorder and Associated Disabilities Interview Schedule-IV (AUDADIS-IV), a computerized, fully structured instrument designed for experienced nonclinician interviewers.³⁶

PTSD was diagnosed when respondents endorsed at least 1 symptom within Criterion B, at least 3 within Criterion C, and at least 2 within Criterion D, lasting at least 1 month (Criterion E), *subsequent to* the worst event they experienced that involved intense fear, helplessness, or horror, and the belief that they or someone close to them might die or be seriously injured or permanently disabled. Diagnoses of PTSD also required that the *DSM-IV* clinical significance criterion of impairment or distress (Criterion F) be met. Respondents with PTSD occurring the year preceding the Wave 2 interview were considered as having a current PTSD, the time frame used by the AUDADIS-IV when assessing the presence of "current" symptoms. A participant with a

12-month *DSM-IV* diagnosis of P1SD was defined as seeking treatment if she/he "went anywhere/had seen anyone to get help with reactions to stressful events" in the previous 12 months. Test-retest reliability of PTSD diagnoses was good (κ =0.64).³⁸ Reliability (κ >0.74) was good to excellent for substance use disorders and fair to good for mood and other anxiety disorders (κ =0.40–0.64).

Measures

Exclusion criteria commonly used in pharmacologic and psychotherapy clinical trials for PTSD were applied to a sample representative of the general US population to evaluate the proportion of participants with a *DSM-IV* diagnosis of PTSD that would be eligible for a typical pharmacologic and psychotherapy clinical trial. The same criteria were applied to a subsample of participants seeking treatment for PTSD to investigate potential differences in eligibility between treatment-seeking and non-treatmentseeking individuals.

We collected the exclusion criteria from clinical trials for PTSD included in a recent meta-analysis³⁹ examining the effects of pharmacologic treatments on PTSD and extracted the psychotherapy clinical trials from a recent meta-analysis⁴⁰ of the efficacy of treatments for PTSD. All the 37 trials used in the pharmacologic meta-analysis and the 56 trials used in the psychotherapy meta-analysis were included in the present study. Two coders (S.F. and N.H.) independently collected all exclusion criteria from the 37 pharmacologic trials and the 56 psychotherapy trials (Table 1). Interrater reliability was adequate (intraclass correlation coefficient = 0.87) and disagreement was resolved by consensus. The median number of exclusion criteria was 10 in the pharmacologic trials (of which 8 could be operationalized) and 4 in the psychotherapy trials (of which all could be operationalized). To estimate the representativeness of typical pharmacologic and psychotherapy clinical trials with traditional exclusion criteria, we applied respectively the 10 and the 4 most commonly used criteria to all individuals with a DSM-IV diagnosis of PTSD within the previous 12 months to the NESARC sample (Tables 2 and 3).

We operationalized exclusion criteria to be consistent with prior work.^{10,20} The criteria "current major depressive episode," "lifetime bipolar disorder," "current drug abuse or dependence," and "current alcohol abuse or dependence" were diagnosed following DSM-IV criteria. The criterion "significant risk of suicide" was considered met if the person reported a suicide attempt during the previous year. The criterion "current psychotic features" was assessed by a single question: "Did a doctor or other health professional ever diagnose you with schizophrenia or psychotic illness or episode during the last year?" The criterion "pregnancy or lactating" was assessed by the following question: "Were you pregnant during the past year?" The criterion "significant medical condition" was approximated using several questions on 12-month angina pectoris, myocardial infarction, or any other form of heart disease, diabetes mellitus, stroke, or cirrhosis or hepatic disease and was considered met if the

<u>is illegal to post this copyrighted PDF on any websi</u>

Table 1. Exclusion Criteria in 37 Pharmacologic and 56 Psychotherapy Clinical Trials Examining the Respective Effects of Treatments on Posttraumatic Stress Disorder (PTSD)^a

	Percentage a Studies Using th		
	Pharmacologic Trials	Psychotherapy Trials	Comparing the Frequency
Exclusion Criteria Present in More Than 10% of	(no.=37),	(no. = 56),	of the Use of Each Criterion
Pharmacologic Trials or Psychotherapy Trials or Both	% (no.)	% (no.)	χ² (<i>P</i> Value)
Current drug abuse or dependence	81.1% (30)	64.3% (36)	2.3 (.13)
Current alcohol abuse or dependence	81.1% (30)	64.3% (36)	2.3 (.13)
Current psychotic features	70.3% (26)	84.0% (47)	1.7 (.19)
Lifetime bipolar disorder	67.6% (25)	25.0% (14)	14.9 (<.01)
Significant medical conditions	62.2% (23)	10.7% (6)	25.1 (<.01)
Currently taking psychotropic medication	51.4% (19)	21.4% (12)	7.7 (<.01)
Current major depressive episode	40.5% (15)	5.4% (3)	15.5 (<.01)
Significant risk of suicide	40.5% (15)	55.4% (31)	1.41 (.24)
Currently receiving psychotherapy	40.5% (15)	14.3% (8)	6.9 (<.01)
Neurologic diseases	32.4% (12)	30.4% (17)	0.05 (.83)
Pregnancy or lactating	43.2% (16)		na
Current panic disorder	35.1% (13)		na
Current generalized anxiety disorder	29.7% (11)		na
Current specific phobia	29.7% (11)		na
Current social anxiety disorder	27.0% (10)		na
Current dysthymia	21.6% (8)		na
Current obsessive-compulsive disorder	21.6% (8)		na
Mental retardation		16.1% (9)	na
Ongoing domestic violence		14.3% (8)	na
History of any dissociative disorder		12.5% (7)	na
Any personality disorder		10.7% (6)	na
^a Boldface type indicates statistical significance.			

Abbreviation: na = not applicable. Symbol: ... = criteria not used or used in less than 10% of trials.

diagnosis was confirmed by a physician. Information to approximate the criteria "currently taking any psychotropic medication," "currently receiving psychotherapy," and "neurological disease" was not available in NESARC.

Analysis Methods

We first determined the percentages (and their 95% confidence intervals [CI]) of survey participants with a current DSM-IV diagnosis of PTSD who would have been excluded by individually applying each exclusion criterion in pharmacologic and psychotherapy clinical trials for PTSD. Because individuals might have been excluded by more than 1 criterion, we also calculated the overall percentage of subjects who would have been excluded by the simultaneous application of all criteria. We conducted these analyses for all participants with a 12-month DSM-IV diagnosis of PTSD (n = 1,715), and for the subsample of individuals who sought treatment (n = 366), in trials examining the effects of pharmacologic and psychological treatments on PTSD separately. Weighted prevalence estimates and their 95% CIs were computed using SUDAAN 10 (RTI International, Research Triangle Park, North Carolina) to account for the complex sampling design of the NESARC.

RESULTS

Exclusion Criteria

Several exclusion criteria were applied in pharmacologic and psychotherapy clinical trials, ie, current drug abuse or dependence, current alcohol abuse or dependence, current psychotic features, lifetime bipolar disorder, significant medical conditions, currently taking psychotropic medication, current major depressive episode, significant risk of suicide, currently receiving psychotherapy, and neurologic disease (Table 1). Among these criteria, 5 were significantly more commonly used in pharmacologic trials than in psychotherapy clinical trials, ie, lifetime bipolar disorder, significant medical conditions, currently taking psychotropic medications, current major depressive episode, and currently receiving psychotherapy. Frequency of use of other criteria did not differ statistically between pharmacologic and psychotherapy trials. In addition, several criteria were typically applied only in pharmacologic trials (ie, current panic disorder, current generalized anxiety disorder, current specific phobia, current social anxiety disorder, current dysthymia, current obsessivecompulsive disorder, and pregnancy or lactating) or only in psychotherapy trials (ie, history of any dissociative disorder, any personality disorder, mental retardation, and ongoing domestic violence). The median number of exclusion criteria used in pharmacologic trials was higher than in psychotherapy trials for PTSD (10 vs 4).

Participants With 12-Month PTSD

Of the 1,715 participants who met *DSM-IV* criteria for PTSD during the previous year, 63.8% (95% CI, 60.8%–66.6%) would have been excluded by at least 1 of the 8 most common and operationalizable criteria in pharmacologic efficacy trials (Table 2). By contrast, the percentage of exclusion by at least 1 of the 4 most common and operationalized criteria in psychotherapy efficacy trials was 19.1% (95% CI, 16.8%–21.7%) (Table 3). Franco et al **It is illega** to post this copyrighted PDF on any website. Table 2. Pharmacologic Treatments for PTSD: Estimated Percentage of Adults

With Posttraumatic Stress Disorder (PTSD) in Wave 2 NESARC Excluded From Typical Clinical Trials by Traditional Exclusion Criteria

			Treatm	hent-Seeking	
	12-Month PTSD (n = 1,715)		Su	Subsample (n=366)	
			(1		
Exclusion Criteria ^a	% ^b	95% Cl	% ^b	95% CI	
Current drug abuse or dependence	6.89	5.41-8.73	7.98	5.24-11.95	
Current alcohol abuse or dependence	13.23	11.36-15.35	16.02	12.00-21.05	
Current psychotic features	3.50	2.51-4.87	8.48	5.65-12.56	
Lifetime bipolar disorder	25.68	23.05-28.50	33.41	27.67-39.68	
Significant medical conditions	22.37	20.06-24.86	25.60	20.54-31.41	
Currently taking psychotropic medication	NA			NA	
Pregnancy or lactating	3.95	2.96-5.27	2.14	0.98-4.60	
Current major depressive episode	18.94	16.82-21.26	26.11	21.21-31.68	
Significant risk of suicide	0.88	0.41-1.87	1.58	0.57-4.29	
Currently receiving psychotherapy	NA			NA	
Exclusion by at least 1 criterion	63.75	60.84-66.57	75.10	69.50-79.97	

^aDerived from the review of 37 clinical trials.

^bPercentages are weighted values. Criteria are ranked by their frequency in typical clinical trials of pharmacologic treatments for PTSD.

Abbreviations: NA = not available in NESARC, NESARC = National Epidemiologic Survey on Alcohol and Related Conditions.

Table 3. Psychotherapy Treatments for PTSD: Estimated Percentage of Adults With Posttraumatic Stress Disorder (PTSD) in Wave 2 NESARC Excluded From Typical Clinical Trials by Traditional Exclusion Criteria

	12-Month PTSD (n=1,715)		Treatment-Seeking Subsample (n = 366)	
Exclusion Criteria ^a	% ^b	95% CI	% ^b	95% CI
Current psychotic features	3.50	2.51-4.87	8.48	5.65-12.56
Current drug abuse or dependence	6.89	5.41-8.73	7.98	5.24-11.95
Current alcohol abuse or dependence	13.23	11.36-15.35	16.02	12.00-21.05
Significant risk of suicide	0.88	0.41-1.87	1.58	0.57-4.29
Exclusion by at least 1 criterion	19.11	16.76-21.70	27.04	21.81-33.00

^aDerived from the review of 56 clinical trials.

^bPercentages are weighted values. Criteria are ranked by their frequency in typical clinical trials of psychotherapeutic treatments for PTSD.

Abbreviation: NESARC = National Epidemiologic Survey on Alcohol and Related Conditions.

In pharmacologic efficacy trials, the percentage of PTSD participants excluded by a single criterion ranged from 0.9% ("significant risk of suicide") to 25.7% ("lifetime bipolar disorder"). In psychotherapy efficacy trials, this percentage ranged from 0.9% ("significant risk of suicide") to 13.2% ("current alcohol abuse or dependence"). In pharmacologic trials, lifetime bipolar disorder, significant medical conditions, and current major depressive episode were the criteria that excluded the greatest proportion of participants with PTSD, whereas having a significant risk of suicide excluded the greatest number of individuals. In psychotherapy trials, having drug and alcohol abuse or dependence excluded the greatest number of individuals with PTSD, whereas the criterion "significant risk of suicide" led to the lowest exclusion rate among these participants.

Treatment-Seeking Subsample

Among the 366 participants seeking treatment, the overall exclusion rate was slightly greater, rising to 75.1% (95% CI, 69.5%–80.0%) in pharmacologic efficacy trials and 27% (95% CI, 21.8%–33.0%) in psychotherapy efficacy

trials. The percentage of participants excluded by a single criterion ranged from 1.6% ("significant risk of suicide") in pharmacologic and psychotherapy trials to 33.4% ("lifetime bipolar disorder") in pharmacologic trials and 16.0% ("current alcohol abuse") in psychotherapy trials.

DISCUSSION

Our study shows that in a typical pharmacologic trial, more than 6 of 10 adults with PTSD in the full sample and more than 7 of 10 respondents in the treatment-seeking subsample would have been excluded by 1 exclusion criterion or more. In contrast, we found that approximately 2 of 10 participants in the full sample and 3 of 10 in the treatment-seeking subsample would have been excluded from participation in a typical psychotherapy trial for PTSD.

These findings indicate that traditional criteria used in pharmacologic trials for PTSD tend to exclude the majority of individuals from participation. A lifetime history of bipolar disorder, significant medical conditions, and current major depressive episode explained a large **It is illegal to post this copy** proportion of inelgibility in pharmacologic trials for PTSD. These exclusion rates in PTSD pharmacologic trials were consistent with previous generalizability studies in adult samples examining the representativeness of clinical trials for a wide range of disorders,^{1,7,10,13,22} in which 60% of the target populations was found to be ineligible. Our result suggests that pharmacologic clinical trials for PTSD may often have limited external validity since they tend to recruit samples of "pure" patients with an uncomplicated diagnosis of PTSD rather than "typical" patients who generally suffer from additional disorders. Pharmacologic clinical trial results for PTSD may thus not be readily generalizable either to community samples or to treatment-seeking populations.

By contrast, we found that psychotherapy efficacy trials for PTSD would typically include most individuals with PTSD with an exclusion rate of approximately 20% in the general population with PTSD and approximately 30% in the treatment-seeking subsample. These estimations rates are in line with those of a recent meta-analysis of psychotherapy trials for PTSD²⁷ that found that clinical trials included roughly 70% of patients referred for treatment and are lower than those reported in psychotherapy trials for other disorders such as generalized anxiety disorder,¹³ social anxiety disorder,¹⁸ and borderline personality disorder¹; in those trials a majority of the target population was found to be excluded. From a public health perspective, this result suggests that efficacy observed in psychotherapy trials for PTSD may reasonably reflect the "true" effect of psychotherapy treatments for PTSD in the general population of individuals with PTSD, although further evidence may be desirable, particularly regarding individuals with comorbid PTSD and substance use disorders who typically are absent from these trials.

The higher overall exclusion rate found in pharmacologic than in psychotherapy trials may be explained by the lower number of exclusion criteria applied and the lower use of criteria excluding participants with comorbid Axis I disorder. Investigators conducting pharmacologic trials may seek to restrict their target populations to decrease potential health hazards for their patients. While some exclusion criteria are necessary in pharmacologic trials to ensure the safety of the study participants (eg, pregnancy and lactation, significant medical conditions, avoiding dangerous interactions of medications), others may be only perpetuated by convention in pharmacologic trials (eg, having a current history of major depressive episode or substance use disorders) and can dramatically narrow the population of eligible participants and thereby negatively impact on external validity. The use of narrow exclusion criteria may sometimes be viewed as a way to support the specificity of the intervention for PTSD. However, psychiatric disorders, which are highly comorbid, are increasingly conceptualized as manifestations of broad liabilities rather than discrete entities,41-43 and several evidence-based, integrated psychotherapies have already been developed for the treatment of comorbid PTSD/ MDD,⁴⁴⁻⁴⁶ possibly explaining the global lower use of exclusion criteria in psychotherapy than in pharmacologic trials for PTSD. Integrated forms of pharmacotherapy that target commonly co-occurring disorders may hold promise for a greater number of individuals with mental disorders and may favor greater inclusive rates of patients with PTSD.^{44,47}

Applying exclusion criteria to the treatment-seeking subsample excluded substantially more participants in pharmacologic and psychotherapy trials for PTSD. As previously suggested,⁴⁸ this result supports that individuals seeking treatment may present greater illness severity and more psychiatric and medical comorbidities, which may possibly lead to higher perceived need of care.⁴⁹ Paradoxically, clinical trials may tend to exclude preferentially individuals who have the greatest overall disease severity and therefore the greatest need for treatment. More inclusive eligibility criteria (particularly with the inclusion of individuals with multiple psychiatric comorbidities) would yield more informative results for mental health care providers and research funding agencies.^{1,50}

This study has several limitations. First, we adopted specific conventions to apply a standard set of exclusion criteria to the NESARC sample. We considered exclusion criteria from 37 pharmacologic and 56 psychotherapy clinical trials for PTSD included in 2 recent meta-analyses.^{27,39} Other conventions might have yielded different exclusion estimates. For example, the 12-month timeframe used in the AUDADIS-IV assessment of "current" symptoms could have led to an overestimation of the exclusion rates. On the other hand, 2 of the traditional exclusion criteria from pharmacologic clinical trials (ie, "currently taking any psychotropic medication" and "currently receiving psychotherapy") could not be operationalized using the NESARC and may have led to an underestimation of participants excluded from pharmacologic clinical trials. Second, participants with current PTSD who sought help for reactions to stressful events during the year preceding the interview were classified as treatmentseeking for PTSD. This definition may have included some participants seeking help for other disorders. Finally, our approach focused on the a priori eligibility of participants and was based on national epidemiologic data. It provides no information on individuals who actually enter those studies. In fact, a substantial proportion of eligible individuals may be unwilling to participate. However, the proportion of individuals that would have been eligible for psychotherapy efficacy trials for PTSD found in this study was consistent with that reported in a multidimensional meta-analysis of psychotherapy for PTSD in which they assessed the participant inclusion rate of those screened for participation.27

CONCLUSION

Although the published pharmacologic clinical trials for PTSD provide a solid base for the provision of evidencebased treatments for PTSD, their external validity is often

Franco et al

It is illegal to post this copy limited since they tend to exclude a majority of participants, particularly those seeking treatment. In contrast, psychotherapy trial results may be applied to most patients with PTSD in routine clinical practice, although further evidence may be desirable, particularly regarding individuals with comorbid PTSD and substance use disorders.

Future pharmacologic clinical trials should carefully consider the trade-offs between the application of each exclusion criterion and its impact on representativeness.^{10,19} Specification a priori of the goals of the study, better justification for each exclusion criterion, and estimation of

Submitted: April 17, 2015; accepted August 5, 2015.

Online first: July 5, 2016.

Potential conflicts of interest: Dr Limosin is a member of the speakers/advisory boards for AstraZeneca, Euthérapie, Janssen, Lundbeck, and Otsuka. The other authors report no conflicts of interest.

Funding/support: Work on this manuscript was supported by National Institutes of Health grants MH076051 and MH082773 (Dr Blanco) and the New York State Psychiatric Institute (Drs Blanco and Hoertel).

Role of the sponsor: The sponsors had no additional role in the design and conduct of the study; collection, management, analysis, and interpretation of the data; and preparation, review, or approval of the manuscript.

Disclaimer: The views and opinions expressed in this study are those of the authors and should not be construed to represent the views of any of the sponsoring organizations, agencies, or the US government.

Additional information: The National Epidemiologic Survey on Alcohol and Related Conditions (NESARC) was sponsored by the National Institute on Alcohol Abuse and Alcoholism (NIAAA) and funded, in part, by the Intramural Program, NIAAA, and National Institutes of Health. The original data set for the NESARC is available from the NIAAA (http:// www.niaaa.nih.gov).

REFERENCES

- Hoertel N, López S, Wang S, et al. Generalizability of pharmacological and psychotherapy clinical trial results for borderline personality disorder to community samples. *Pers Disord*. 2015;6(1):81–87.
- 2. Velasco E. *Inclusion Criteria*. Thousand Oaks, CA: SAGE Publications, Inc; 2010.
- 3. Humphreys K, Weisner C. Use of exclusion criteria in selecting research subjects and its effect on the generalizability of alcohol treatment outcome studies. *Am J Psychiatry*. 2000;157(4):588–594.
- March JS, Silva SG, Compton S, et al. The case for practical clinical trials in psychiatry. *Am J Psychiatry*. 2005;162(5):836–846.
- Dzewaltówski DA, Estabrooks PA, Klesges LM, et al. Behavior change intervention research in community settings: how generalizable are the results? *Health Promot Int*. 2004;19(2):235–245.
- Hoertel N, Le Strat Y, Lavaud P, et al. Generalizability of clinical trial results for bipolar disorder to community samples: findings from the National Epidemiologic Survey on Alcohol and Related Conditions. *J Clin Psychiatry*. 2013;74(3):265–270.
- 7. Hoertel N, Le Strat Y, De Maricourt P, et al. Are subjects in treatment trials of panic disorder

representative of patients in routine clinical practice? results from a national sample. *J Affect Disord*. 2013;146(3):383–389.

agencies.41

- Hoertel N, Le Strat Y, Limosin F, et al. Prevalence of subthreshold hypomania and impact on internal validity of RCTs for major depressive disorder: results from a national epidemiological sample. *PLoS ONE*. 2013;8(2):e55448.
- Gaynes BN, Warden D, Trivedi MH, et al. What did STAR*D teach us? results from a large-scale, practical, clinical trial for patients with depression. *Psychiatr Serv*. 2009;60(11):1439–1445.
- Blanco C, Olfson M, Goodwin RD, et al. Generalizability of clinical trial results for major depression to community samples: results from the National Epidemiologic Survey on Alcohol and Related Conditions. J Clin Psychiatry. 2008;69(8):1276–1280.
- Zimmerman M, Chelminski I, Posternak MA. Generalizability of antidepressant efficacy trials: differences between depressed psychiatric outpatients who would or would not qualify for an efficacy trial. *Am J Psychiatry*. 2005;162(7):1370–1372.
- Zimmerman M, Mattia JI, Posternak MA. Are subjects in pharmacological treatment trials of depression representative of patients in routine clinical practice? *Am J Psychiatry*. 2002;159(3):469–473.
- Hoertel N, Le Strat Y, Blanco C, et al. Generalizability of clinical trial results for generalized anxiety disorder to community samples. *Depress Anxiety*. 2012;29(7):614–620.
- Humphreys K. A review of the impact of exclusion criteria on the generalizability of schizophrenia treatment research [published online ahead of print June 20, 2014]. *Clin Schizophr Relat Psychoses.*
- Khan AY, Preskorn SH, Baker B. Effect of study criteria on recruitment and generalizability of the results. J Clin Psychopharmacol. 2005;25(3):271–275.
- Leucht S, Heres S, Hamann J, et al. Methodological issues in current antipsychotic drug trials. Schizophr Bull. 2008;34(2):275–285.
- Boter H, Derks EM, Fleischhacker WW, et all; EUFEST Study Group. Generalizability of the results of efficacy trials in first-episode schizophrenia: comparisons between subgroups of participants of the European First Episode Schizophrenia Trial (EUFEST). J Clin Psychiatry. 2010;71(1):58–65.
- Hoertel N, de Maricourt P, Katz J, et al. Are participants in pharmacological and psychotherapy treatment trials for social anxiety disorder representative of patients in real-life settings? J Clin Psychopharmacol. 2014;34(6):697–703.
- Hoertel N, Falissard B, Humphreys K, et al. Do clinical trials of treatment of alcohol dependence adequately enroll participants

with co-occurring independent mood and anxiety disorders? an analysis of data from the National Epidemiologic Survey on Alcohol and Related Conditions (NESARC). *J Clin Psychiatry*. 2014;75(3):231–237.

 Blanco C, Olfson M, Okuda M, et al. Generalizability of clinical trials for alcohol dependence to community samples. *Drug Alcohol Depend*. 2008;98(1–2):123–128.

the proportion of individuals ineligible for the trial wou

assist study design. Although efficacy studies may benefit

from relatively stringent eligibility criteria to maximize

detection of drug-placebo differences, effectiveness studies

could place a larger emphasis on the generalizability of

their findings by imposing less stringent eligibility criteria.

Developing integrated forms of pharmacotherapy and

psychotherapy that simultaneously target commonly

overlapping psychiatric disorders may yield more informative

results for mental health care providers and research funding

- Le Strat Y, Rehm J, Le Foll B. How generalisable to community samples are clinical trial results for treatment of nicotine dependence: a comparison of common eligibility criteria with respondents of a large representative general population survey. *Tob Control.* 2011;20(5):338–343.
- Okuda M, Hasin DS, Olfson M, et al. Generalizability of clinical trials for cannabis dependence to community samples. *Drug Alcohol Depend*. 2010;111(1–2):177–181.
- Stein DJ, Ipser J, McAnda N. Pharmacotherapy of posttraumatic stress disorder: a review of meta-analyses and treatment guidelines. CNS Spectr. 2009;14(suppl 1):25–31.
- Cukor J, Spitalnick J, Difede J, et al. Emerging treatments for PTSD. *Clin Psychol Rev.* 2009;29(8):715–726.
- Foa EB. Psychosocial treatment of posttraumatic stress disorder. J Clin Psychiatry. 2000;61(suppl 5):43–48, discussion 49–51.
- Bisson JI, Roberts NP, Andrew M, et al. Psychological therapies for chronic posttraumatic stress disorder (PTSD) in adults. Cochrane Database Syst Rev. 2013;12:CD003388.
- Bradley R, Greene J, Russ E, et al. A multidimensional meta-analysis of psychotherapy for PTSD. *Am J Psychiatry*. 2005;162(2):214–227.
- Pacella ML, Hruska B, Delahanty DL. The physical health consequences of PTSD and PTSD symptoms: a meta-analytic review. *J Anxiety Disord*. 2013;27(1):33–46.
- Blanco C, Xu Y, Brady K, et al. Comorbidity of posttraumatic stress disorder with alcohol dependence among US adults: results from National Epidemiological Survey on Alcohol and Related Conditions. Drug Alcohol Depend. 2013;132(3):630–638.
- Brady KT, Killeen TK, Brewerton T, et al. Comorbidity of psychiatric disorders and posttraumatic stress disorder. J Clin Psychiatry. 2000;61(suppl 7):22–32.
- McCall-Hosenfeld JS, Mukherjee S, Lehman EB. The prevalence and correlates of lifetime psychiatric disorders and trauma exposures in urban and rural settings: results from the national comorbidity survey replication (NCS-R). *PLoS ONE*. 2014;9(11):e112416.
- Breslau N. The epidemiology of posttraumatic stress disorder: what is the extent of the problem? *J Clin Psychiatry*. 2001;62(suppl 17):16–22.
- 33. Breslau N. The epidemiology of trauma, PTSD,

e980 PSYCHIATRIST.COM

J Clin Psychiatry 77:8, August 2016

and other posttrauma disorders. Trauma Ruan WJ, Goldstein RB, Chou SP, et al. The

Violence Abuse. 2009;10(3):198-210.

- 34. Pietrzak RH, Goldstein RB, Southwick SM, et al. Prevalence and Axis I comorbidity of full and partial posttraumatic stress disorder in the United States: results from Wave 2 of the National Epidemiologic Survey on Alcohol and Related Conditions. J Anxiety Disord. 2011:25(3):456-465.
- 35. Grant BF, Stinson FS, Dawson DA, et al. Prevalence and co-occurrence of substance use disorders and independent mood and anxiety disorders: results from the National Epidemiologic Survey on Alcohol and Related Conditions. Arch Gen Psychiatry. 2004;61(8):807-816.
- 36. Grant BF, Harford TC, Dawson DA, et al. The Alcohol Use Disorder and Associated Disabilities Interview schedule (AUDADIS): reliability of alcohol and drug modules in a general population sample. Drug Alcohol Depend, 1995;39(1):37-44.
- 37. Grant BF, Goldstein RB, Chou SP, et al. Sociodemographic and psychopathologic predictors of first incidence of DSM-IV substance use, mood and anxiety disorders: results from the Wave 2 National Epidemiologic Survey on Alcohol and Related Conditions. Mol Psychiatry. 2009;14(11):1051-1066.

alcohol use disorder and associated disabilities interview schedule-IV (AUDADIS-IV): reliability of new psychiatric diagnostic modules and risk factors in a general population sample. Drug Alcohol Depend. 2008;92(1-3):27-36.

- 39. Ipser JC, Stein DJ. Evidence-based pharmacotherapy of post-traumatic stress disorder (PTSD). Int J Neuropsychopharmacol. 2012;15(6):825-840.
- 40. Watts BV, Schnurr PP, Mayo L, et al. Metaanalysis of the efficacy of treatments for posttraumatic stress disorder. J Clin Psychiatry. 2013;74(6):e541-e550.
- 41. Hoertel N, Franco S, Wall MM, et al. Mental disorders and risk of suicide attempt: a national prospective study. Mol Psychiatry. 2015;20(6):718-726.
- 42. Blanco C, Krueger RF, Hasin DS, et al. Mapping common psychiatric disorders: structure and predictive validity in the national epidemiologic survey on alcohol and related conditions. JAMA Psychiatry. 2013;70(2):199-208.
- 43. Hoertel N. McMahon K. Olfson M. et al. A dimensional liability model of age differences in mental disorder prevalence: evidence from a national sample. J Psychiatr Res. 2015:64:107-113.
- 44. Mulick PS, Naugle AE. Behavioral activation for

comorbid PTSD and major depression: a case study. Cognit Behav Pract. 2004;11(4):378-387.

- 45. Wagner AW, Zatzick DF, Ghesquiere A, et al. Behavioral activation as an early intervention for posttraumatic stress disorder and depression among physically injured trauma survivors. Cognit Behav Pract. 2007;14(4):341-349.
- 46. Jiang RF, Tong HO, Delucchi KL, et al. Interpersonal psychotherapy versus treatment as usual for PTSD and depression among Sichuan earthquake survivors: a randomized clinical trial. Confl Health. 2014;8(1):14.
- 47. Capone C, Eaton E, McGrath AC, et al. Integrated Cognitive Behavioral Therapy (ICBT) For PTSD and substance use in Iraq and Afghanistan veterans: a feasibility study. J Trauma Stress Disord Treat. 2014;3(4).
- 48. Cohen P, Cohen J. The clinician's illusion. Arch Gen Psychiatry. 1984;41(12):1178-1182.
- 49 Kirchner JE, Booth BM, Owen RR, et al. Predictors of patient entry into alcohol treatment after initial diagnosis. J Behav Health Serv Res. 2000;27(3):339-346.
- 50. Wisniewski SR, Rush AJ, Nierenberg AA, et al. Can phase III trial results of antidepressant medications be generalized to clinical practice? a STAR*D report. Am J Psychiatry. 2009;166(5):599-607.